\begin{table}%ta.1
\caption{\label{tab:tab_1}Data relative to Trojans and Plutinos.} 
%\centerline
 {\begin{tabular}{ccccc}
\hline\hline
 {Name} &  {Sample} &  {Class$^a$} &  {$B-R^b$} &  {$H_R$$^c$} \\
\hline
2001QR322 &ST06 &1:1 &1.26 &7.67  \\  
2004UP10  &ST06 &1:1 &1.16 &8.50  \\ 
2005TN53  &ST06 &1:1 &1.29 &8.89  \\ 
2005TO74  &ST06 &1:1 &1.34 &8.29  \\ 
Pluto     &JL01   &3:2 &1.34 &--1.37  \\  
1993RO    &TR00   &3:2 &1.36 &8.41   \\ 
1993SB    &TR00   &3:2 &1.29 &7.68   \\ 
1993SC    &TR98/RT99   &3:2 &1.97 &6.53   \\  
1994JR1   &M2S99   &3:2 &1.61 &7.06   \\  
1994TB    &TR98/RT99   &3:2 &1.78 &7.43   \\  
1995HM5   &TR98/RT99   &3:2 &1.01 &7.88   \\ 
1995QY9   &M2S99   &3:2 &1.21 &7.02   \\  
1995QZ9   &TR00   &3:2 &1.40 &8.06   \\  
1996RR20  &TR00   &3:2 &1.87 &6.49   \\ 
1996SZ4   &TR00   &3:2 &1.35 &7.92   \\ 
1996TP66  &TR98/RT99   &3:2 &1.85 &6.71   \\  
1996TQ66  &TR98/RT99   &3:2 &1.86 &6.99   \\ 
1997QJ4   &LP02  &3:2 &1.10 &7.84   \\  
1998HK151 &M2S02   &3:2 &1.24 &6.78   \\  
1998UR43  &MBOSS    &3:2 &1.35 &8.09   \\  
1998US43  &LP04  &3:2 &1.19 &7.75   \\  
1998VG44  &TRC07   &3:2 &1.52 &6.10   \\  
1998WS31  &LP04  &3:2 &1.31 &7.77   \\  
1998WU31  &LP04  &3:2 &1.23 &7.99   \\ 
1998WV31  &LP04  &3:2 &1.34 &7.53   \\ 
1998WW24  &LP04  &3:2 &1.35 &7.84   \\ 
1998WZ31  &LP04  &3:2 &1.26 &7.93   \\ 
1999TC36  &TRC03   &3:2 &1.74 &4.64   \\  
1999TR11  &TR00   &3:2 &1.77 &7.88   \\  
2000EB173 &TR03   &3:2 &1.60 &4.43   \\ 
2000GN171 &TRC07   &3:2 &1.57 &5.62   \\
2001KB77  &TRC07   &3:2 &1.39 &7.18   \\  
2001KD77  &LP04  &3:2 &1.75 &5.74   \\  
2001KX76  &M2S02   &3:2 &1.64 &3.25   \\  
2001KY76  &M2S05   &3:2 &1.85 &6.68   \\ 
2001QF298 &TRC07   &3:2 &1.14 &4.91   \\ 
2002GF32  &M2S05   &3:2 &1.76 &5.95   \\ 
2002GV32  &M2S05   &3:2 &1.96 &6.75   \\ 
2002VE95  &TRC07   &3:2 &1.79 &5.06   \\ 
2002VR128 &TRC07   &3:2 &1.54 &4.83   \\ 
2002XV93  &TRC07   &3:2 &1.09 &4.36   \\ 
2003AZ84  &TRC07   &3:2 &1.06 &3.46   \\ 
2003VS2   &TRC07   &3:2 &1.52 &4.14   \\ 
2004DW    &TRC07   &3:2 &1.05 &1.92   \\ 
2004EW95  &TRC07   &3:2 &1.08 &6.08   \\ 
\hline
\end{tabular}}
\par
\smallskip
$^a$ {Resonance with Neptune.}
$^b$ {Color index.}
$^c$ {$R$-filter absolute magnitude.}\protect\\
Samples -- [TR98/RT99]: {\cite{1998Natur.392...49T}, \cite{1999Natur.398..129R}};
[TR00]: {\cite{2000Natur.407..979T}};
[TR03]: {\cite{2003Icar..161..181T}};
[TRC03]: {\cite{2003ApJ...599L..49T}};
[TRC07]: {Database: \protect\url{http://www.physics.nau.edu/~tegler/research/survey.htm}};
[LP02]: {\cite{2002A&A...395..297B}}; 
[LP04]: {\cite{2004Icar..170..153P}};
%[LP04 LP2n]: {\cite{2004Icar..170..153P} (NTT data)};
[MBOSS]: {Database: \cite{2002A&A...389..641H}, \cite{2001A&A...380..347D}, \cite{2001Icar..152..246G}};
[M2S99]: {\cite{1999Icar..142..476B}};
%[M2S00 MS2]: {\cite{2000AJ....120..496B}};
%[M2S01 MS3]: {\cite{2001Icar..154..277D}};
[M2S02]: {\cite{2002AJ....124.2279D}};
[M2S05]: {\cite{2005Icar..174...90D}};
[JL01]: {\cite{2001AJ....122.2099J}};
[ST06]: {\cite{2006Sci...313..511S}}.
\end{table}